April 20, 2022

Prof. Panteleimon Giannakopoulos  
Editor-in-Chief  
*World Journal of Psychiatry*

Dear Prof. Giannakopoulos:

This letter is in reference to our revised manuscript (76436) entitled, “A cross-sectional survey followed longitudinal study on mental health and insomnia of people related with sporadic COVID-19 case” that we have revised and would like you to reconsider for publication in *World Journal of Psychiatry*. Below are detailed responses to the reviewers’ comments. Specific changes in the revised text are highlighted in **bold** font. We thank you and the reviewers for the thoughtful comments.

I have reviewed the Peer-Review Report, full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the *World Journal of Psychiatry*, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. Please provide decomposable Figures (in which all components are movable and editable), organize them into a single PowerPoint file. Please authors are required to provide standard three-line tables, that is, only the top line, bottom line, and column line are displayed, while other table lines are hidden. The contents of each cell in the table should conform to the editing specifications, and the lines of each row or column of the table should be aligned. Do not use carriage returns or spaces to replace lines or vertical lines and do not segment cell content.

**Response:** We appreciate the Editor’s positive evaluation. We revised figure 1 and Table1 - 3 according to your requirements. Please see line 217-317, line 290-261, 302-303, 336-337, respectively.

Reviewer #1:

- **Scientific Quality:** Grade A (Excellent)
- **Language Quality:** Grade A (Priority publishing)
- **Conclusion:** Accept (High priority)

**Specific Comments to Authors:** this manuscript present good topic, to clarify the severity of psychological problems and insomnia of medical staff in and community residents around a hospital with a sporadic COVID-19 case reported. I have only one comment for this article: 1. The figure 1 is not that clear, consider to use image with good resolution.

**Response:** We appreciate the reviewer’s positive evaluation. The figure 1 was revised. Please see line 217-237.
Reviewer #2:

**Scientific Quality:** Grade B (Very good)

**Language Quality:** Grade B (Minor language polishing)

**Conclusion:** Accept (General priority)

**Specific Comments to Authors:** The manuscript has been promoted during revision.

**Response:** We appreciate the reviewer's positive evaluation.